Author's response to reviews

Title: Cost savings of reduced constipation rates attributed to increased dietary fiber intakes: a decision-analytic model

Authors:

Jordana K Schmier (jschmier@exponent.com)
Paige E Miller (pmiller@exponent.com)
Jessica A Levine (jlevine@exponent.com)
Vanessa Perez (vperez@exponent.com)
Kevin C Maki (kevin.maki@mxns.com)
Tia M Rains (tia.rains@mxns.com)
Latha Devareddy (latha.devareddy@kellogg.com)
Lisa M Sanders (lisa.sanders@kellogg.com)
Dominik D Alexander (dalexander@exponent.com)

Version: 2 Date: 9 April 2014

Author's response to reviews:

Our cover letter followed by a point-by-point response to the reviewers is attached.